\documentclass[a4paper,twoside,12pt]{article}
%\usepackage [reqno] {amsmath}
\usepackage{amsfonts,amstext,amssymb}
\usepackage{amsmath}
\usepackage{german}
\usepackage{graphicx}
\usepackage{fullpage}

\newcommand{\ZETTELNUMMER}{8a}
\newcommand{\ABGABEDATUM}{on 20. June 2016 in the tutorial session}

\newcounter{AUFGNR}
\setcounter{AUFGNR}{1}
\newcommand{\AUFGABE}[2]{\vspace{0.3cm}\item[Problem~\arabic{AUFGNR}]\stepcounter{AUFGNR} #1\hfill\emph{#2}}


\newcommand{\floor}[1]{\left\lfloor{#1}\right\rfloor}
\newcommand{\ceil}[1]{\left\lceil{#1}\right\rceil}
\newcommand{\half}[1]{\frac{#1}{2}}
\newcommand{\eps}{\varepsilon}
\newcommand{\N}{\mathbb{N}}



\renewcommand{\labelenumi}{(\alph{enumi})}
\renewcommand{\labelenumii}{(\roman{enumii})}


\begin{document}
\pagestyle{empty}
\hrule\medskip
\rule{0ex}{0ex}\\[-1ex]
\ZETTELNUMMER. Assignment on 

\smallskip
\noindent
\large
\textbf{Computational Complexity Theory}\hfill SoSe 2016 \\[0.5ex]
\normalsize
Wolfgang Mulzer, Claudia Dieckmann

\medskip\hrule

\smallskip
\noindent
\textbf{Due} \ABGABEDATUM

\vskip 0.5cm


\begin{description}
\AUFGABE {Artistic Expression} {10 points}

Draw a picture that reflects your current knowledge of complexity theory.
Your drawing should feature at least ten complexity classes, eight complete 
problems, and four colors. In the tutorial session on Monday, 20.06.2016,
you will have an opportunity to share your artistic rendition with the
group.

\end{description}
\end{document}
